Author’s response to reviews

Title: Traumatic penile amputation: A case report

Authors:

Tushar Patial (drtusharpatial@gmail.com)
Girish Sharma (shred.tushar@gmail.com)
Pamposh Raina (pamposhraina@mail.com)

Version: 1 Date: 06 Jan 2017

Author’s response to reviews:

Answers to comments

1. As your manuscript describes a case report, we require a statement regarding consent for publication. We cannot currently accept the sentence "Given by the father of the patient". As per our submission guidelines (https://bmcurology.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report)

A: We have obtained written informed consent from the patients father for publication. The same has now been mentioned in the declarations section.

2. Please consider having the text edited a native English speaking colleague. There are issues with grammar, wording, and punctuation that will prevent effective peer review of the current version.

A: The entire article has been scrutinized thoroughly, and any errors of grammar, wording and punctuation may kindly be pardoned.

3. Please provide all of the information we require in the Declarations section, as per our submission guidelines - https://bmcurology.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report

Currently, some of the sections are incomplete, or the statements contained are insufficient.

A: All necessary changes have been made.